Author's response to reviews

Title: Misdiagnosed murine typhus in a patient with multiple cerebral infarctions and hemorrhage: A case report

Authors:

Ziqi Xu Dr. (zyxuziqi@zju.edu.cn)
Xiongchao Zhu Dr. (sunnydays163@163.com)
Qunying Lu Dr. (lqycdc@163.com)
Xia Li Dr. (lix992@yahoo.com.cn)
Yewen Hu Dr. (zysnhu5101@163.com)

Version: 6 Date: 1 July 2015

Author's response to reviews:

Dear Editors:

Thank you giving us a chance again to revise our research article entitled “Misdiagnosed murine typhus in a patient with multiple cerebral infarctions and hemorrhage: A case report” for consideration of publication at your prestigious journal.

This article had been revised after full consideration of the comments of experienced editors. The responses are follows.

1. Please include a statement to specify whether informed consent for the publication of this case report was obtained. For all manuscripts that include details, images, or videos relating to individual participants, written informed consent for the publication of these must be obtained from the participants (or their parent or legal guardian in the case of children under 16) and a statement to this effect should appear in the manuscript. If the participant has died, then consent for publication must be sought from the next of kin of the participant. You can use our consent form to obtain consent for publication from the participant(s), or a consent form from your own institution or region if you prefer. This documentation must be made available to Editors on request, and will be treated confidentially.

The statement to specify whether informed consent for the publication of this case report was obtained and was added to the paper. (see line152-154)

2. Please move all tables from the additional files into the main manuscript file, and remove all versions of the manuscript from the additional files.

We have removed all table and all versions of the manuscript from the additional files.